Author's response to reviews

Title: Effects of interval between age at first birth and age at diagnosis on breast cancer survival according to menopausal status: A register-based study in Korea

Authors:

   JungSun Lee (gsjslee@gmail.com)
   Minkyung Oh (minkyung@busanpaik.ac.kr)

Version: 3 Date: 11 March 2014

Author's response to reviews:

Dear reviewer

Thank you for nice and immediate response about this manuscript. We all changed the manuscript according to editorial requirement, such as ethical approval, abbreviations, author's contributions, and tables.